Author's response to reviews

Title: The effect of health promoting resources on work participation in population reporting musculoskeletal pain. A sample from the Nord-Trondelag Health Study, HUNT 3, Norway.

Authors:

Heidi Sivertsen (heidisiv@online.no)
Geir Arild Espnes (Geir.Arild.Espnes@svt.ntnu.no)

Version: 2 Date: 16 February 2012

Author's response to reviews:

Since first submission authors' contribution section is completed with "HS designed the study, participated in the drafting of the manuscript, data analysis, and interpretation of results. GEA critical revised, and approved the final manuscript" and is placed after the competing interests section.